Distribution Agreement

In presenting this thesis or dissertation as a partial fulfillment of the requirements for an advanced degree from Emory University, I hereby grant to Emory University and its agents the non-exclusive license to archive, make accessible, and display my thesis or dissertation in whole or in part in all forms of media, now or hereafter known, including display on the world wide web. I understand that I may select some access restrictions as part of the online submission of this thesis or dissertation. I retain all ownership rights to the copyright of the thesis or dissertation. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

Sonia Kapil

April 21, 2020

Maternal Anthropometry and Current Nutritional Interventions to Prevent Adverse Birth Outcomes in the Humanitarian Context: A Systematic Review

By

Sonia Kapil Master of Public Health

Environmental Health

Jeremy Sarnat, ScD Committee Chair

Mija-Tesse Ververs, MMed, MPH, RD Committee Member

Maternal Anthropometry and Current Nutritional Interventions to Prevent Adverse Birth Outcomes in the Humanitarian Context: A Systematic Review

By

Sonia Kapil

B.S. University of Pittsburgh 2020

Thesis Committee Chair: Jeremy Sarnat, ScD

An abstract of a thesis submitted to the Faculty of the Rollins School of Public Health of Emory University in partial fulfillment of the requirements for the degree of Master of Public Health in Environmental Health 2022

Abstract

Maternal Anthropometry and Current Nutritional Interventions to Prevent Adverse Birth Outcomes in the Humanitarian Context: A Systematic Review

By Sonia Kapil

Background: Maternal undernutrition in the humanitarian setting is a prominent concern that leaves both mothers and newborns at-risk for adverse outcomes. These risks range from maternal mortality, low birth weight, intra-uterine growth restriction, pre-term birth, small-for-gestational-age, and stunting at birth. In 2013, Médecins Sans Frontiéres (MSF) Switzerland analyzed the relationship of maternal anthropometric indicators for acute malnutrition with adverse birth outcomes. Mid-upper-arm-circumference (MUAC) was established as the preferential indicator, with a proposed cut-off value of < 23 cm as the criteria for enrollment of pregnant women in nutritional programs.

Objectives: The primary goals of this systematic review were to provide an update to the MSF review by (1) determining if MUAC remains the preferential indicator to identify LBW and other potential risks in mothers and their children; (2) determining what specific anthropometric cutoffs have been used to identify adverse birth outcomes and enroll pregnant women in nutritional programs; (3) determining whether or not these nutritional programs, contingent upon enrollment based on anthropometry, are successful in preventing adverse birth or maternal outcomes.

Methods: Two literature reviews covering September 2012 to February 2021 were conducted on the topics of maternal anthropometry to identify adverse birth or maternal outcomes and maternal anthropometry and nutritional interventions. Adapted Quality Assessments for individual studies were completed.

Results: MUAC was demonstrated as a proxy for a BMI of $< 18.5 \text{ kg/m}^2$ to detect undernutrition. A maternal MUAC threshold value of < 23 cm was found to be strongly predictive for identification of pregnant women as at-risk for adverse birth outcomes. Nutritional interventions with enrollment based on MUAC values defined as undernourished that demonstrated improvements in nutritional status were limited, but included ready-to-use-supplementary food and food-based balanced energy and protein supplementation.

Conclusion: The research analyzed in this systematic review supports maternal MUAC as an alternative and more feasible measurement to BMI for identifying pregnant women as undernourished and in need of nutritional intervention to prevent adverse birth outcomes in the humanitarian setting. The preferential MUAC indicator in this context is < 23 cm. Research on nutritional interventions with enrollment contingent upon anthropometry must be further studied.

Maternal Anthropometry and Current Nutritional Interventions to Prevent Adverse Birth Outcomes in the Humanitarian Context: A Systematic Review

By

Sonia Kapil

B.S. University of Pittsburgh 2020

Thesis Committee Chair: Jeremy Sarnat, ScD

A thesis submitted to the Faculty of the Rollins School of Public Health of Emory University in partial fulfillment of the requirements for the degree of Master of Public Health in Environmental Health 2022

Acknowledgements

I would like to express my deepest and humblest gratitude to Dr. Mija Ververs for your invaluable mentorship, guidance, advice, and support during this process. Your vast knowledge, insight, and thoughtful feedback, has provided me a tremendous learning opportunity. I would like to thank Dr. Jeremy Sarnat for serving as my Thesis Committee Chair and being an amazing mentor within the department. You have provided me with so many opportunities throughout these past two years at Rollins and I cannot thank you enough for your time and words of wisdom as an advisor and professor.

A final note to thank all of my family and friends for their love, support, and encouragement. Thank you, Neil, for having the patience to help me get through many late night editing sessions and meticulously going over my work. Thank you, Nanu, for your everlasting encouragement and for being my biggest believer and role model. A special thank you to Mom & Dad for getting me to where I am today. Mom, your warm words and endless love mean the world to me. Dad, I am honored to follow in your footsteps throughout my academic endeavors; your hard work and ambition continues to serve as source of inspiration.

INTRODUCTION	1
Background	1
Purpose Statement	5
Significance Statement	6
METHODS	7
Search Strategy	7
Study Eligibility Criteria: Inclusion Criteria	7
Study Eligibility Criteria: Exclusion Criteria	8
Study Eligibility Criteria: Outcome Measures	8
Data Synthesis	9
Quality Assessment	9
RESULTS	11
Studies Identified	11
Maternal BMI and Adverse Outcomes	15
Relationship Between MUAC and BMI	17
MUAC Cut-off Threshold	19
Applying Nutritional Interventions Based on Anthropometry	22
Additional Potential Nutritional Interventions	25
DISCUSSION	27
Anthropometric Indicators	27
Nutritional Programs	29
Limitations	30
Recommendations	30
CONCLUSION	31
APPENDICES	32
Appendix A. Adapted Newcastle Ottawa Quality Assessment Scale: Cross-Sectional Studies	32
Appendix B. Adapted Newcastle Ottawa Quality Assessment Scale: Cohort Studies	34
Appendix C. Adapted Newcastle Ottawa Quality Assessment Scale: Case-Control Studies	36
Appendix D. Adapted Joanna Briggs Institute Critical Appraisal Checklist: Randomized Controlled Trials	38
Appendix E. Adapted AMSTAR Checklist: Systematic Reviews and Meta-Analyses	40
REFERENCES	42

Table of Contents

List of Tables and Figures

Table 1. Common Maternal Anthropometric Indicators	2
Table 2. Nutritional Programs to Address Maternal Malnutrition	4
Table 3. Definitions of Outcome Measures	9
Table 4. Studies Post-September 2012 Looking at Maternal BMI and Adverse Outcomes	16
Table 5. Studies Post-September 2012 Looking at the Relationship Between MUAC and BMI	18
Table 6. Studies Post-September 2012 Using Maternal MUAC to Identify Adverse Birth Outcomes	21
Table 7. Studies Post-September 2012 Looking at the Application of Nutritional Interventions Based on Maternal	
Anthropometry	24
Table 8. Studies Post-September 2012 Looking at the Additional Potential Nutritional Interventions	26

Figure 1. Flow Diagram Illustrating Article Review Process: Anthropometry to Identify Adverse Outcomes	12
Figure 2. Flow Diagram Illustrating Article Review Process: Anthropometry and Nutritional Interventions	13
Figure 3. Countries of Studies Involving Anthropometry to Identify Adverse Outcomes	14
Figure 4. Countries of Studies Involving Anthropometry and Nutritional Interventions	14

Acronyms

BEP	Balanced Energy and Protein
BMI	Body-Mass Index
CSB+	Corn-Soy Blend
ENN	Emergency Nutrition Network
IUGR	Intra-Uterine Growth Restriction
LBW	Low Birth Weight
IFA	Iron and Folic Acid
MMN	Multiple Micronutrient
MSF	Médecins Sans Frontiéres
MUAC	Mid-Upper-Arm Circumference
РТВ	Pre-Term Birth
PW	Pregnant Women
RUSF	Ready-To-Use Supplementary Food
SFP	Supplementary Feeding Program
SGA	Small-For-Gestational-Age
SQ-LNS	Small-Quantity Lipid-Based Nutrient Supplementation
UNICEF	United Nations Children's Fund
UNIMMAP	United Nations International Multiple Micronutrient Antenatal
	Preparation
WHO	World Health Organization

INTRODUCTION

Background

Maternal undernutrition is a prominent global health concern leaving both mothers and newborns vulnerable to adverse health outcomes. Undernutrition is defined as the deficiency of essential nutrients and the inadequate intake of dietary energy¹. Undernutrition can result in malnutrition, which is a condition that develops from the lack of sufficient vitamins, minerals, and specific nutrients required to maintain healthy tissue and organ function². Malnutrition is the broad term for poor nutrition that can result from either undernutrition or overnutrition³. Acute malnutrition is the form of undernutrition that results from inadequate energy or protein intake from nutritional deficiency⁴.

In 2013, Médecins Sans Frontiéres (MSF) Switzerland undertook a literature review of articles published between January 1995 and September 2012 covering anthropometric indicators that can identify pregnant women (PW) as acutely malnourished and at-risk for adverse birth outcomes. The focus was on the humanitarian setting, with an emphasis on the African and Asian contexts where many humanitarian emergencies occur⁵. Acute malnutrition in pregnant women in these settings of conflict or natural disaster is prevalent because of the lack of health care availability and food accessibility within their environments, often leading to health risks for both the mother and child⁶. These risks include maternal mortality, low birth weight (LBW), intra-uterine growth restriction (IUGR), pre-term birth (PTB), small-for-gestational-age (SGA) newborns, and stunting at birth.

After an extensive search, MSF concluded that maternal mid-upper-arm circumference (MUAC) can be used as a reliable indicator for risk of low birth weight. *Table 1* explains the differences between the various anthropometric measurements commonly used in reference to

maternal health. Maternal MUAC is the preferential indicator, as opposed to body mass index (BMI), maternal weight for gestational age, maternal weight gain, or maternal stature because of the strong association it has with birth weight, the simplicity of the measurement in the humanitarian context, and its independence from gestational age. The proposed conservative cutoff value to enroll pregnant women in nutritional programs, such as supplementary feeding programs (SFP), is a MUAC of < 23 cm⁵.

Table 1. Common Maternal Anthropometric Indicators							
Measurement	Description						
MUAC	Singular measurement of the circumference of PW's arm at the midpoint between the shoulder and elbow ⁷						
BMI	Calculation on the measure of body fat based on two measurements: height and weight ⁸						
Weight for gestational age	Single measurement of weight at specific time points during pregnancy; knowledge of gestational age necessary						
Weight gain	Calculated difference in weight from two points during pregnancy; knowledge of gestational age not necessary						
Stature	Single measurement of height; non-modifiable in adult PW ⁹						

Birth weight is a prominent barometer of fetal health, with LBW serving as an indication for risk of infant morbidity and mortality¹⁰. The prevention of poor birth outcomes, predominantly LBW, is being explored through nutritional programs with the inclusion of various types of nutrient supplementations. As recommended by the World Health Organization (WHO), the standard practice of care for maternal and fetus well-being to prevent LBW and PTB includes the use of daily iron-folic acid (IFA) supplementation¹¹. Newly researched nutritional programs that address the issue of maternal undernutrition in efforts to improve maternal and birth outcomes include the utilization of fortified corn-soy blend (CSB+)¹², United Nations multiple micronutrient preparation (UNIMMAP)¹³, lipid-based ready-to-use supplemental food (RUSF)¹³, maternal balanced energy protein (BEP)¹⁴ supplementation, antenatal multiple micronutrient (MMN)¹⁵ supplementation, and preventative small-quantity lipid-based nutrient supplementation (SQ-LNS)¹⁶. *Table 2* describes each of these nutritional programs. Effective nutritional programs will compensate for potential macro- and micro-nutrient deficiencies and lead to a reduction in maternal and child undernutrition.

Nutritional Program	Description	Contents						
Iron-folic acid (IFA) supplementation ¹⁷	Recommended for all pregnant women, regardless of nutritional status (standard practice of care)	Elemental iron and folic acid						
Fortified corn-soy blend (CSB+) ¹⁸	Used as complementary food throughout the world by USAID implementing partners with an intended use of increasing nutrient values and caloric density	Cooked blend of milled, heat-treated corn with soybeans fortified with a vitamin and mineral premix						
United Nations multiple micronutrient preparation (UNIMMAP) ¹⁹	Formulated specifically for PW in emergency situations and research studies	Contains 15 vitamins and minerals at dosages that approximate the recommended dietary allowances (RDA) for pregnancy						
Lipid-based ready-to-use supplemental food (RUSF) ²⁰	Medium-quantity lipid-based nutrient supplements with micronutrient-enriching paste designed to treat moderate acute malnutrition, providing 50-100% of energy needed	Heat treated seeds/pulses/cereals, sugar, milk powder, vegetable oils, vitamin, and minerals						
Maternal balanced energy protein (BEP) supplementation ^{21, 22}	Proteins energy supplementation that provides less than 25% of the total energy content to fill nutrient gaps while not displacing food intake in PW	Lipid-based peanut paste and vanilla biscuit						
Antenatal multiple micronutrient supplementation (MMN) ²³	Intended to fill nutritional gaps assumed to be common during pregnancy	~1 recommended dietary allowance (RDA) of vitamins and minerals						
Preventative small-quantity lipid-based nutrient supplementation (SQ-LNS) ²⁴	Fortified lipid-based paste/spread providing less than 50% of energy needed	Heat treated peanut/pulses/cereals, milk powder, vegetable oils, sugar, maltodextrin, vitamins, and minerals (23 micronutrients)						

Table 2. Nutritional Programs to Address Maternal Malnutrition

Purpose Statement

A recommendation from the MSF review includes determining the extent to which enrollment in nutritional programs based on a maternal MUAC of < 23 cm averts the risk of LBW infants. An emphasis was also placed on evaluating if MUAC remains the favored anthropometric indicator or if the combined use of easily measurable anthropometric indicators is preferable for predicting the risk of adverse birth outcomes, specifically in the humanitarian context⁵.

The purpose of this systematic review is to incorporate these recommendations through an analysis of studies published post-September 2012 by (1) determining if MUAC remains the preferential indicator to identify LBW and other potential risks in mothers and their children; (2) determining what specific anthropometric cut-offs have been used to identify adverse birth outcomes and enroll pregnant women in nutritional programs; (3) determining whether or not these nutritional programs, contingent upon enrollment based on anthropometry, are successful in preventing adverse birth or maternal outcomes.

Significance Statement

This study addresses the vulnerability of pregnant women living in emergency settings in the African and Asian context. The prevention of poor birth outcomes is vital for the health of both the mother and newborn child. Identifying a practical anthropometric measurement and a cut-off point to identify pregnant women as undernourished in this particular setting can assist the implementation of necessary interventions to avoid unfavorable outcomes. Pregnant women in complex protracted crises often have an unknown exact gestational age, may not have access to healthcare facilities, and may maintain an inadequate nutritional status. Often, these pregnant women will not be seen again by a healthcare provider until birth, so a quick identification process to determine if they need immediate nutritional assistance will aid in the prevention of adverse effects.

A key gap in women's maternal nutrition is that there is no agreed upon standard set in the Sphere Handbook that defines acute malnutrition through an optimal, context-specific MUAC cut-off point²⁵. The Emergency Nutrition Network's (ENN) 2022 Technical Briefing Paper on Women's Nutrition²⁶ recommends a clarification of the indicators of anthropometric status to be linked to nutritional program decision-making. This research is intended to assist necessary international guidance regarding maternal MUAC cut-off point guidelines and the formulation of nutritional programs based on these guidelines through the amalgamation of the presented current literature results.

METHODS

This study hypothesizes that in the humanitarian context, a MUAC of < 23 cm remains the preferential anthropometric indicator and cut-off point to identify adverse outcomes in mothers and newborns. Additionally, it is postulated that nutritional programs that incorporate this threshold as criteria for enrollment are effective at preventing LBW, IUGR, PTB, and SGA.

Search Strategy

This study is an analysis of data abstracted from two comprehensive literature searches conducted primarily in the PubMed and Embase electronic databases. Additional eligible studies were sought after reviewing the reference lists of identified articles. (1) The first literature search focused on anthropometry, with a priority on maternal MUAC, to identify risks of adverse birth outcomes. (2) The second literature search focused on using anthropometry to enroll pregnant women in nutritional interventions to prevent adverse birth or maternal outcomes.

Key terms:

 (1) maternal anthropometry, pregnancy, mid-upper-arm-circumference, adverse birth outcomes, low birth weight, maternal outcomes
 (2) maternal anthropometry, pregnancy, nutritional interventions, nutritional programs, adverse birth outcomes

Study Eligibility Criteria: Inclusion Criteria

The PRISMA guidelines were used to aid article and study selection. *Figure 1* displays search results from the (1) first literature search and *Figure 2* displays search results from the (2) second literature search. The dates covered for these searches in both databases were September

2012 through February 2021. The inclusion criteria to identify relevant studies include those available in full-text, peer-reviewed, available in English, and focused on adult maternal anthropometry; an exception was made regarding two studies relating BMI and MUAC, which involved non-pregnant adult women^{39, 40}. This review was not specifically restricted to studies done in low- and middle-income countries or protracted humanitarian emergency settings. Publications with cross-sectional, retrospective cohort, prospective cohort, unmatched case-control, longitudinal, randomized control, evaluation, and any relevant systematic reviews or meta-analysis study designs were included.

Study Eligibility Criteria: Exclusion Criteria

For both literature searches, duplicate publications and studies analyzing the same study populations for similar outcomes were excluded. Additional exclusions from the literature review searches comprised of results involving or focusing on the following subjects: twins, triplets, adolescents, substance abuse, anemia, cigarette smoking, in-vitro fertilization, drugs and hormones, disease, and obesity. Further restrictions were placed on studies without full-text available and non-English and non-human studies.

Study Eligibility Criteria: Outcome Measures

The outcome measures for the studies from the first literature review focused on both maternal and newborn birth outcomes. These included LBW, IUGR, PTB, SGA, and stunted at birth. Definitions are presented in *Table 3*.

Table 3. Definitions of Outcome Measures							
Outcome	Definition						
LBW	Newborn whose birth weight is $< 2,500 \text{ g}^{27}$						
IUGR	Fetus whose estimated birth weight is below the 10^{th} percentile for its gestational age, birth weight < 2,500 g for gestational age greater than 37 weeks, and abdominal circumference below the 2.5 th percentile ²⁸						
РТВ	Newborn whose birth is before 37 weeks of gestation ²⁹						
SGA	Newborn whose weight below the 10 th percentile for its gestational age ³⁰						
Stunted at Birth	Impaired linear birth; a newborn who falls below -2 standard deviations on the recommended length/height-for-age growth charts ³¹						

Data Synthesis

Since individual studies were not comparable, different approaches were taken for study analyses, and the data was limited, a meta-analysis was not conducted. Data was synthesized based on the results of each individual study, and quantitative results were extracted to be organized in thematic tables.

Quality Assessment

Studies that were determined to fit the indicated inclusion criteria were assessed by one reviewer on their strength, based on study type. According to individual study scores, each study can be categorized as good quality, fair quality, or poor quality. The Newcastle-Ottawa Quality Assessment Scale was adapted based on study type and used to assess and score the quality of selected, cross-sectional studies (*Appendix A*), cohort studies (*Appendix B*), and case-control studies (*Appendix C*) based on three domains: Selection, Comparability, and Outcome/Exposure³². The Joanna Briggs Institute Critical Appraisal Checklist³³ was adapted and used to assess and score the quality of selected randomized controlled trial studies (*Appendix D*). The AMSTAR Checklist³⁴ was adapted and used to assess and score the quality of select systematic reviews and meta-analyses (*Appendix E*).

RESULTS

Studies Identified

Figure 1 displays a flow diagram summarizing the systematic review process of literature selection for assessment for (1) Anthropometry to Identify Adverse Outcomes. A total of 5,099 articles were initially identified, with an additional 25 articles found through investigation of the reference lists of initially identified articles. These articles were sorted through and narrowed down to 39 relevant full-text articles. After complete screening and eligibility review, a total of 13 of articles were included in this systematic review. The Quality Assessments determined all included studies to be categorized as either good or fair quality. The studies included were conducted in the following countries: Bangladesh, Brazil, Cambodia, China, Ethiopia, India, Senegal, and South Africa (*Figure 3*).

Figure 2 displays a flow diagram summarizing the systematic review process of literature selection for assessment for (2) Anthropometry and Nutritional Interventions. A total of 109 articles were initially identified, with an additional 11 articles found through investigation of the reference lists of initially identified articles. These articles were sorted through and narrowed down to 11 relevant full-text articles. After complete screening and eligibility review, a total of 8 articles were included in this systematic review. The Quality Assessments determined all included studies to be categorized as either good or fair quality. The studies included were conducted in the following countries: Bangladesh, Burkina Faso, Ethiopia, Ghana, Indonesia, and Malawi (*Figure 4*).

Figure 1. Flow Diagram Illustrating Article Review Process: Anthropometry to Identify Adverse Outcomes



Figure 2. Flow Diagram Illustrating Article Review Process: Anthropometry and Nutritional Interventions





Figure 3. Countries of Studies Involving Anthropometry to Identify Adverse Outcomes

Figure 4. Countries of Studies Involving Anthropometry and Nutritional Interventions



Maternal BMI and Adverse Outcomes

The BMI cutoff point of $< 18.5 \text{ kg/m}^2$ has been demonstrated to increase the risk of LBW, IUGR, and SGA, as shown in *Table 4*. The study looking at the relationship between maternal BMI and the outcome of LBW was a cross-sectional study conducted in urban and rural households in selected countries in Africa. It was not specified when in the pregnancy BMI measurements were taken and calculated. The association between maternal BMI and LBW was found to be statistically significant only for Senegal (OR = 1.961, 95% CI: 1.259, 3.055); however, in Burkina Faso, Malawi, and Uganda, underweight mothers - defined by a BMI of <18.5 kg/m² – were also found to have a higher likelihood of LBW newborns³⁵. In the South Gondar Zone of Ethiopia, a cross-sectional study was carried out in four hospitals where maternal BMI was measured at delivery for all pregnant women delivering in selected hospitals. Here a BMI of $< 18.5 \text{ kg/m}^2$ was found to be a significant predictor of the outcome IUGR (aOR = 2.57, 95% CI: 1.72, 3.83)³⁶. The study from Southwest China analyzed data from pregnant women enrolled in a randomized controlled trial also defined underweight BMI as $< 18.5 \text{ kg/m}^2$. BMI was measured at the first visit during early pregnancy, which was defined as 11 to 14 weeks of gestation. Early pregnancy BMI classified as underweight was found to have a statistically significant association with increased risk for SGA newborns (Chinese BMI category: OR = 2.08, 95% CI: 1.17, 3.70; WHO Asian BMI category: OR = 2.04, 95% CI: 1.14, 3.66; WHO European BMI category: OR = 2.10, 95% CI: 1.18, 3.72). Early pregnancy BMI classified as underweight was not found to have a statistically significant association with increased risk for LBW newborns (Chinese BMI category: OR = 1.03, 95% CI: 0.20, 5.35; WHO Asian BMI category: OR = 1.11, 95% CI: 0.21, 5.81; WHO European BMI category: OR = 1.14, 95% CI: $(0.22, 5.98)^{37}$.

Table 4. Studies Post-September 2012 Looking at Maternal BMI and Adverse Outcomes											
Study	Country	Study Population	Sample Size (n)	Study type	Time of Measurement	Cut-off Value	Statistical Test	LBW	IUGR	SGA	
He et al. 2018 ³⁵	Senegal	PW of urban and rural households	1665 women	Cross- sectional	Not specified	< 18.5 kg/m^2	OR (95% CI)	1.961 (1.259, 3.055)			
Tesfa et al. 2020 ³⁶	Ethiopia	PW delivering in a hospital	803 women	Cross- sectional	Delivery	< 18.5 kg/m^2	OR (95% CI)		2.57 (1.72, 3.83)		
Chen et al. 2021 ³⁷	China	PW attending antenatal services	1273 women	Randomized controlled trial	Early pregnancy (11- 14 weeks of gestation)	< 18.5 kg/m^2	OR (95% CI)	* Chinese BMI category: 1.03 (0.20, 5.35,); ** WHO Asian BMI category: 1.11 (0.21, 5.81); *** WHO European BMI category: 1.14 (0.22, 5.98)		* Chinese BMI category: 2.08 (1.17, 3.70); ** WHO Asian BMI category: 2.04 (1.14, 3.66); *** WHO European BMI category: 2.10 (1.18, 3.72)	

* Chinese BMI category: underweight: <18.5 kg/m2, normal weight: 18.5–23.9 kg/m2, overweight: 24.0–27.9 kg/m2 and obese: >28.0 kg/m2
** WHO Asian BMI category: underweight: <18.5 kg/m2, normal weight: 18.5–22.9 kg/m2, overweight: 23.0–24.9 kg/m2 and obese: >25.0 kg/m2
*** WHO European BMI category: underweight: <18.5 kg/m2, normal weight: 18.5–24.9 kg/m2, overweight: 25–29.9 kg/m2 and obese: >30 kg/m2
Statistically significant values are in bold

Relationship Between MUAC and BMI

Five studies post-September 2012 were identified demonstrating MUAC as a functional surrogate for BMI. *Table 5* demonstrates the relationship between BMI and MUAC measurements. All studies here utilize a BMI measurement of $< 18.5 \text{ kg/m}^2$, which has been established by the WHO as indicating an adult within the underweight range and an adult pregnant woman as at-risk for adverse birth outcomes³⁸. Two of these studies had a study population consisting of non-pregnant adult women, but were included because of the relevance of evaluating the relationship between BMI and MUAC. From these individual cross-sectional studies conducted in Bangladesh and India, MUAC measurements of < 23.9 cm (rounded to < 24 cm)³⁹ and $< 23.2 \text{ cm}^{40}$, respectively, were found to be correlated to a BMI of $< 18.5 \text{ kg/m}^2$.

From the three studies specifically looking at populations consisting of pregnant women, three different MUAC values were all found to have a relationship with being underweight, as indicated with a BMI of < 18.5 kg/m². In the cross-sectional study from South Africa, a MUAC of < 22.8 cm (rounded to < 23 cm) was found to correlate strongly (r = 0.92, p < 0.0001) with BMI in pregnant women up to 30 weeks' gestation⁴¹. In another cross-sectional study from India, it was found that a MUAC of < 23.5 cm is significantly associated with BMI of < 18.5 kg/m² (aOR = 7.91, 95% CI: 4.27, 14.65) during the first trimester of pregnancy⁴². In the retrospective cohort study from Brazil, a MUAC of < 25.75 cm was correlated with a BMI of < 18.5 kg/m² (r = 0.872, p < 0.0001) during weeks 19 to 21 of pregnancy⁴³.

Table 5. Studies Post-September 2012 Looking at the Relationship Between MUAC and BMI										
Study	Country	Study Population	Sample Size (n)	Study Type	Time of Measurement	BMI Measurement	MUAC Measurement	Statistical test Results		
Sultana et al. 2015 ³⁹	Bangladesh	Non-pregnant adult women	650 women	Cross- sectional	N/A	< 18.5 kg/m^2	< 23.9cm (rounded to < 24 cm)	Sensitivity of 92.6% and Specificity of 76.64%; Pearson Correlation, r = 0.828 (p < 0.001)		
Fakier et al. 2017 ⁴¹	South Africa	PW attending maternity services	164 women	Cross- sectional	< 30 weeks of gestation	< 18.5 kg/m^2	< 22.8 cm (rounded to < 23 cm)	Correlation, r = 0.92 (p < 0.0001)		
Kumar et al. 2019 ⁴⁰	India	Non-pregnant adult women	1716 women	Cross- sectional	N/A	< 18.5 kg/m^2	< 23.2 cm	Correlation, r = 0.860 (95% CI: (0.831, 0.883); p < 0.001)		
Mishra et al. 2020 ⁴²	India	PW attending maternity services	440 women	Cross- sectional	1st trimester	< 18.5 kg/m^2	< 23.5 cm	aOR = 7.91 (4.27– 14.65); Correlation, r = 0.57 (p < 0.001)		
Miele et al. 2021 ⁴³	Brazil	PW attending prenatal care services	1165 women	Retrospective cohort	Three set points: 19-21 weeks, 27-29 weeks, 37-39 weeks	< 18.5 kg/m^2	< 25.75 cm (19- 21 weeks)	Correlation, r = 0.872 (p < 0.0001)		

MUAC Cut-off Threshold

The five studies in *Table 6* demonstrate the specific maternal MUAC cut-off threshold values and the corresponding birth outcomes of LBW, IUGR, SGA, and stunted at birth. Three of these studies indicated a MUAC of < 23 cm as strongly predictive for identifying pregnant women as at-risk for at least one of these adverse outcomes, while one study used a MUAC cut-off value of \leq 23 cm and another study used < 22 cm. None of these cut-off values were found to be linked to gestational age.

Three studies looked at the adverse birth outcome of LBW; two studies in Ethiopia and one study in India. The first study from Ethiopia was a cross-sectional facility based study aiming to identify factors associated with LBW. This study found that a MUAC < 23 cm measured at delivery was significantly associated with LBW (aOR = 3.4, 95% CI: 1.38, 8.60)⁴⁴. The second study from Ethiopia was an unmatched case-control study that found that a MUAC < 22 cm measured at delivery was significantly associated with LBW (aOR = 2.89, 95% CI: 1.58, 5.29)⁴⁵. From the prospective cohort study in India, a MUAC of \leq 23 cm measured during the first and second trimester was found to be associated with LBW (OR = 1.083, 95% CI: 0.46, 2.58), though despite this particular association with LBW, this result was not statistically significant at the 0.05 significance level⁴⁶.

Another cross-sectional study in Ethiopia looked at the adverse birth outcome of IUGR. Here, a MUAC < 23 cm measured at delivery was found to be significantly associated with IUGR (aOR = 2.10, 95% CI: 1.39, 3.01)⁴⁷. The prospective cohort study in India looked at the adverse birth outcome of SGA. Here, a MUAC of \leq 23 cm measured during the first and second trimester was found to be associated with SGA (OR = 0.90, 95% CI: 0.42, 1.93), though this result was not statistically significant at the 0.05 significance level⁴⁶. A longitudinal study done in Cambodia looked at the adverse birth outcome of being stunted at birth. Here, a MUAC of < 23 cm measured during the third trimester was found to be associated with being stunted at birth (aOR = 1.621, 95% CI: 0.998, 2.636), although this result was borderline significant with a p-value of 0.051⁴⁸.

Table 6. Studies Post-September 2012 Using Maternal MUAC to Identify Adverse Birth Outcomes											
Study	Country	Study Population	Sample Size (n)	Study type	Time of MUAC Measurement	MUAC Cut-off Value	Statistical Test	LBW	IUGR	SGA	Stunted at Birth
Adane, Dachew 2018 ⁴⁴	Ethiopia	PW delivering in a hospital	662 women	Cross- sectional	Delivery	< 23 cm	OR (95% CI)	3.4 (1.38, 8.60)			
Vasundhara et al. 2019 ⁴⁶	India	PW attending antenatal services	928 women	Prospective cohort	1st and 2nd trimester	≤23cm	OR (95% CI)	1.083 (0.46, 2.58)		0.90 (0.42, 1.93)	
Siyoum, Melese 2019 ⁴⁵	Ethiopia	PW delivering in a hospital	330 women	Unmatched case-control	Delivery	< 22 cm	OR (95% CI)	2.89 (1.58, 5.29)			
Kpewou et al. 2020 ⁴⁸	Cambodia	PW attending antenatal services	779 women	Longitudinal	3rd trimester	< 23cm	OR (95% CI)				1.621 (0.998, 2.636)
Tesfa et al. 2020 ⁴⁷	Ethiopia	PW delivering in a hospital	803 women	Cross- sectional	Delivery	< 23 cm	OR (95% CI)		2.10 (1.39, 3.01)		

Statistically significant values are in bold

Applying Nutritional Interventions Based on Anthropometry

Table 7 demonstrates four studies looking at the application of different nutritional interventions on the basis of anthropometry. In Malawi, a single-blind randomized controlled clinical trial was conducted in pregnant women in their second or third trimester. Moderate malnutrition was defined at enrollment with a MUAC between ≥ 20.6 cm and ≤ 23.0 cm. Pregnant women who fit this categorization received one of three dietary treatment regimens: RUSF, CSB+ with UNIMMAP, or CSB+ with IFA (standard of care). The incidence of LBW infants for the RUSF intervention group was 18%, for the CSB+ with UNIMMAP intervention group was 24% and for the CSB+ with IFA standard of care group was 17% (p = 0.02)⁴⁹.

In Bangladesh, a village-matched evaluation study was conducted in pregnant women in their first or early second trimester. Undernourishment was defined at enrollment as having a MUAC ≤ 22.1 cm. Pregnant women who were identified as undernourished using this definition were either enrolled in the intervention group, which received food-based BEP supplements, or the control group, which did not receive the food-based BEP supplements. The intervention reduced the risk of LBW by 88.58% (RRR = 0.8858)⁵⁰.

For an unmatched case-control study conducted in the Sidama Zone of Ethiopia, cases considered were LBW newborns and controls were healthy newborns (≥ 2500 g). The likelihood of MUAC < 23 cm at delivery was 4.27 times higher among mothers of the cases than the controls, as compared to having a MUAC ≥ 23 cm at delivery (aOR = 4.27, 95% CI: 2.24, 8.12). The odds of taking not IFA during pregnancy was 3.92 times higher in mothers of the cases compared to the controls (aOR = 3.92, 95% CI: 1.80, 8.50)⁵¹. Here, not receiving IFA supplementation and maternal malnutrition indicated by MUAC < 23 cm were found to be independent determinants of birth weight among newborns. In Indonesia, an evaluation study defined malnourishment as having a MUAC < 23.5 cm, where MUAC was measured at two points in the pregnancy, but the timing of measurements was not specified. Malnourished pregnant women received a complementary feeding program in the form of a supplementary feeding biscuit containing 260 kcal of energy, 13 g of fat, and 8 g of protein, and healthy pregnant women did not receive any intervention. Regardless of biscuit dose, the supplementary feeding biscuit did not have an effect on pregnancy outcomes in malnourished pregnant women, including MUAC, gestational weight, and birth weight⁵².

Table 7. Studies Post-September 2012 Looking at the Application of Nutritional Interventions Based on Maternal Anthropometry

Study	Country	Study Population	Sample Size (n)	Study Type	Time of Measurement	Anthropometric value for enrollment	Intervention	Outcomes
Callaghan- Gillespie et al. 2017 ⁴⁹	Malawi	PW attending antenatal clinics	1828 women	Single- blind randomized controlled clinical trial	Enrollment in 2nd and 3rd trimester; measurements taken every 2 weeks after enrollment	Moderate malnutrition defined as MUAC \geq 20.6 cm and \leq 23.0 cm	RUSF; CSB+ with UNIMMAP; or CSB+ with IFA (standard of care)	Incidence of LBW infants: RUSF = 18%; CSB+ with UNIMMAP = 24%; CSB+ with IFA = 17% ($p = 0.02$)
Stevens et al. 2018 ⁵⁰	Bangladesh	Undernourished PW	87 women	Village- matched evaluation ^a	Enrollment in 1st and early 2nd trimester	Undernourished defined as MUAC ≤ 22.1 cm	Food-based balanced protein energy supplementation	Intervention reduced risk of LBW by 88.58% (RRR ^b = 0.8858); NNT ^c = 6.32
Bekela et al. 2020 ⁵¹	Ethiopia	PW delivering in selected public hospitals	354 women	Unmatched case- control	Delivery	N/A	IFA supplementation	Not taking IFA supplementation in pregnancy in cases vs. controls: $OR = 3.92$ (95% $CI = 1.80-8.50$). MUAC < 23 cm in cases vs. controls: 4.27 (2.24, 8.12) (p = 0.001)
Henrick et al. 2020 ⁵²	Indonesia	PW enrolled in complementary feeding program	211 women	Evaluation study	Not specified; measurements taken 2 times in pregnancy (specific timing unclear)	Malnourished defined as MUAC < 23.5 cm	Supplementary feeding biscuit	No effect of the supplementary biscuit on MUAC or gestational weight among malnourished pregnant women

^a the village-matched evaluation used principles of a cluster randomized controlled trial

^bRRR = Relative Risk Reduction

^cNNT = Number Needed to Treat

Additional Potential Nutritional Interventions

Additional potential interventions to address maternal undernourishment include the use of MMNs and SQ-LNS, as shown in *Table 8*. These studies do not use anthropometry as a basis of enrollment, but serve as a demonstration of potential prospective effective interventions to address maternal malnutrition. A randomized controlled study from Malawi had three intervention groups: IFA, MMN, or SQ-LNS. Here, SQ-LNS did not demonstrate improvements in child weight, MUAC, or stunting, when compared to IFA or MMN⁵³. A similar randomized controlled study from Ghana also had the same three intervention groups: IFA, MMN, or SQ-LNS. Here, the SQ-LNS group had a lower prevalence (57.4%) of inadequate GWG than the MMN group (67.2%)⁵⁴. No differences in maternal mortality were found between any of the groups.

A systematic review examined four studies from Bangladesh (LNS compared to IFA), Burkina Faso (LNS compared to MMN), Ghana (LNS compared to both IFA and MMN), and Malawi (LNS compared to both IFA and MMN). Here, LNS had a slight positive effect on birth weight, SGA, and newborn stunting, but no difference in maternal mortality or GWG, when compared to IFA. LNS and MMN did not have a significant difference in maternal and birth outcomes. LNS compared to both IFA and MMN did not have a significant different in maternal and birth outcomes⁵⁵.

A systematic review and meta-analysis evaluating different supplementation interventions on maternal, birth, child health, and developmental outcomes found that MMN compared to IFA improved LBW, SGA, and PTB. LNS compared to MMN slightly reduced the risk of SGA, but had no effect on LBW or PTB.

Table 8. Stu	Table 8. Studies Post-September 2012 Looking at the Additional Potential Nutritional Interventions									
Study	Country	Study Population	Sample Size (n)	Study Type	Intervention	Outcomes				
Ashorn et al. 2015 ⁵³	Malawi	PW attending antenatal clinics	869 women	Randomized controlled trial	IFA, MMN, or SQ- LNS (enrollment at ≤ 20 weeks of gestation)	No effect with SQ-LNS on improving child growth (mean child weight, MUAC, head circumference, prevalence of stunting, and mean length) compared to IFA or MMN				
Adu- Afarwuah et al. 2017 ⁵⁴	Ghana	PW attending antenatal clinics (at ≤ 20 weeks of gestation)	1320 women	Randomized controlled trial	IFA, MMN, or SQ- LNS (enrollment at ≤ 20 weeks of gestation)	SQ-LNS group (57.4% prevalence) had lower prevalence of inadequate GWG than the MMN group (67.2% prevalence) ($p = 0.03$)				
Das et al. 2018 ⁵⁵	Bangladesh, Burkina Faso, Ghana, and Malawi	PW from a total of 4 studies	8018 women, collectively	Systematic review	 (1) LNS compared to IFA; (2) LNS compared to MMN; (3) (4) LNS compared to both IFA and MMN (enrollment timing varies) 	LNS vs. IFA: no maternal outcome differences; LNS had slightly higher mean birth weight, length and reduction in SGA and stunting; no difference in PTB, stillbirth, or neonatal mortality LNS vs. MMN: no maternal outcome differences; no difference in LBW, mean birth weight, length, SGA, PTB, or neonatal mortality				
Oh et al. 2020 ⁵⁶	N/A	PW from a total of 72 studies	451,723 women, collectively	Systematic review and meta-analysis	IFA compared to folic acid; MMN compared to IFA; LNS compared to MMN (enrollment timing varies from < 13 to < 37 weeks of gestation)	MMN vs. IFA: MMN improved LBW, PTB, SGA LNS vs. MMN: LNS slightly reduced risk of SGA, no effect on LBW or PTB				

DISCUSSION

In this systematic review, an exploration of recently published literature was conducted concerning anthropometric indicators that can identify pregnant women as undernourished and at-risk for adverse birth outcomes, as well as nutritional programs and their effectiveness. The purpose was to serve as a follow-up of the research updates presented in the 2013 MSF Switzerland⁵ review by focusing on MUAC as a satisfactory substitute for BMI to identify risks such as LBW, IUGR, PTB, and SGA; establishing the specific anthropometric context-specific cut-off points to identify adverse outcomes and enroll pregnant women in nutritional programs; and analyzing if the nutritional programs are successful in achieving the desired outcome of reducing adverse birth and/or maternal outcomes. All included studies were deemed good or fair quality, with none being categorized as poor quality, according to the adapted Quality Assessments based on specific study type, thus further supporting the recommendations extrapolated from this systematic review.

Anthropometric Indicators

BMI has remained the gold standard for measuring the amount of body fat; however, the question remains whether or not it is the most feasible anthropometric method to determine nutritional status in pregnant women who are living in humanitarian and low-resource settings⁴¹. The BMI cutoff point of < 18.5 kg/m² has been demonstrated to be a sufficient predictor of adverse outcomes, such as LBW³⁵, IUGR³⁶, and SGA³⁷. Studies have found that pre-pregnancy weight and pre-pregnancy BMI may serve as good indicators of risk for adverse birth outcomes⁵⁶; however, these studies were not included in this review because pre-pregnancy

measurements are often unknown or not verifiable, rendering pre-pregnancy weight and prepregnancy BMI to have no significant practical value in this context.

Based on the current findings, maternal MUAC appears to remain the most reliable and practical anthropometric indicator to detect undernutrition, which increases the risk of adverse birth outcomes and the need for nutritional interventions, serving as a proxy for maternal BMI. The two studies that were included that involved non-pregnant adult women for the sake of demonstrating the association between MUAC and BMI^{39, 40} should be considered with care, since maternal body composition changes during pregnancy⁵⁷. While all studies used the BMI cutoff point of < 18.5 kg/m² as the definition of underweight, the MUAC values from studies specifically looking at pregnant women that were associated with underweight BMI ranged between < 22.8 cm to < 25.75 cm.

Since MUAC is established as the preferential indicator over BMI, the next research question focuses on specific MUAC cut-off point(s) to identify pregnant women at-risk for adverse outcomes. All studies in *Table 6* determined maternal MUAC cut-off values to be independent of gestational age, which is particularly important for this context since gestational age is often unknown for pregnant women in humanitarian emergencies. A majority of these studies utilized a MUAC threshold of < 23 cm to identify pregnant women at-risk for the following birth outcomes: LBW^{44, 45, 46}, IUGR⁴⁷, SGA⁴⁶, and stunted at birth⁴⁸. This confirms the use of MUAC as the context specific preferential indicator over BMI.

The study from India that used a maternal MUAC cut-off value of ≤ 23 cm did not demonstrate any significant associations with LBW or SGA. Here, the usage of the less than or equal to symbol (\leq) is unclear since it leaves the specific threshold open for interpretation⁴⁶. Since measurements were not completed in millimeters, but rather in centimeters, it is unclear whether values between 23.1 cm to 23.9 cm are included in this threshold criteria. Additionally, this study states that a MUAC of < 24 cm had poor specificity, but higher sensitivity, so it is recommended that a MUAC of \leq 23 cm be used. Although this study was established as good quality, it should be considered in perspective of other studies that determined a MUAC of < 23 cm as the potential indicator of birth risks.

Nutritional Programs

RUSF was found as a better intervention to address maternal weight gain and prevention of LBW infants, compared to CSB+ with UNIMMAP⁴⁹. BEP could potentially serve as an alternative to RUSF to treat acute malnutrition, with an additional benefit being its sustainability and cost-effectiveness⁵⁰.

Timing of enrollment in nutritional program based on anthropometry is essential. The primary goal is to accurately detect undernutrition in pregnant women as early as possible in pregnancy to implement a nutritional intervention that improves nutritional status and reducing the risk of adverse birth and maternal outcomes. If length of enrollment in the specific nutritional program is not long enough, such as beginning in the third trimester, there is the possibility that any changes in birth or maternal outcomes are not associated or unrelated to the nutritional program itself, but rather attributed to an external factor.

The 2021 Lancet Series on Maternal and Child Undernutrition Progress⁵⁸ reiterated the importance of the effectiveness of antenatal MMNs, along with preventative SQ-LNSs as an emerging intervention that has shown positive effects on childhood growth and low- and middle-income countries. While studies MMN and SQ-LNS supplementations were not using MUAC or any particular anthropometry as enrollment criteria for undernutrition, they may serve as the

starting point for future studies. In the systematic review and meta-analysis presented, MMN did demonstrate improvement in LBW, PTB, and SGA compared to IFA alone⁵⁶.

Limitations

Limitations of the current literature include the lack of research on the outcome of maternal mortality and the limited research on nutritional interventions with enrollment based on anthropometry. Additionally, the context of these studies was not necessarily in humanitarian emergencies or conflict settings, though the results can be applied to this context. In some studies, a less than or equal to sign was indicated, which lacks clarity for exact cut-off thresholds.

Limitations of this particular systematic review are that there was only one Quality Assessment reviewer, which has the potential of introducing bias. Ideally, having at least two independent reviewers would limit this bias and ensure all relevant studies are included in the review. Additionally, the studies are not all comparable since they varied in sample size, methodology, and context. The studies included were limited to availability in English only, which could have filtered out valuable results. Since grey literature is not peer-reviewed, it was not included, which also could have contained noteworthy data.

Recommendations

Future recommendations would be to enroll pregnant women in nutritional interventions based on the well-supported MUAC < 23 cm in efforts to reduce the risk of adverse outcomes. Additionally, these studies primarily focus solely on adverse birth outcomes rather than maternal outcomes. Maternal anthropometry and maternal mortality should be further researched.

CONCLUSION

The current research supports using maternal MUAC as an alternative anthropometric measurement to BMI for identifying pregnant women as acutely malnourished and in need of nutritional intervention to prevent adverse birth outcomes. This is particularly noteworthy in resource-limited settings, such as protracted humanitarian settings or emergencies. An advantage of measuring MUAC is that it requires minimal training and is reliable in identifying nutritional status. Initially, there has been no universal absolute cut-off value identified; however, this review supports the specific cut-off threshold for maternal MUAC in this context as < 23 cm. According to current findings, enrollment in nutritional programs such as RUSF and food-based BEP supplements based on a MUAC value defined as undernourished may address the issue of LBW and other adverse outcomes. In addition to standard of care IFA supplementation, preventative SQ-LNS may serve as a promising intervention to prevent maternal and newborn undernutrition, but more research must be conducted before recommending SQ-LNS as an effective intervention to prevent adverse birth and maternal outcomes.

APPENDICES

Appendix A. Adapted Newcastle Ottawa Quality Assessment Scale: Cross-Sectional Studies

Selection (maximum 3 points)

- 1) Representative: 1 point was given if the sample was truly representative of the target population (underweight or undernourished PW)
- 2) Sample Size: 1 point was given if sample size was justified and satisfactory
- 3) Non-Respondents/Non-Included Subjects: 1 point was given if comparability between respondents and non-respondents characteristics is established and the response rate is satisfactory

Comparability (maximum 2 points)

- 1) Comparability of Subjects: 1 point was given if subjects in different outcome groups are comparable
- 2) Confounding Factors: 1 point was given if study was adjusted for confounding factors

Outcome (maximum 3 points)

- 1) Assessment of Outcome: up to 2 points given if the assessment of the outcome is satisfactory and well-explained
- 2) Statistical Test: 1 point was given if the statistical test used to analyze the data was clearly described and appropriate, and the measure of association was presented, including confidence intervals and the probability level (p-value)

Total Points:

- 7, 8: good quality
- 5, 6: fair quality
- 4 points or less: low quality

Appendix A. Adapted Newcastle Ottawa Quality Assessment Scale: Cross-Sectional Studies

Study	Tesfa et al. 2020	He et al. 2018	Sultana et al. 2015	Fakier et al. 2017	Kumar et al. 2019	Mishra et al. 2020	Adane, Dachew 2018
Selection (maximum 3 points)							
Representative	1	1	0	1	0	1	1
Sample Size	1	1	1	0	1	1	1
Non-Respondents/Non- Included Subjects	1	1	0	1	1	1	1
Comparability (maximum 2 points)							
Comparability of Subjects	1	1	1	1	1	1	1
Confounding Factors	1	1	0	0	0	1	1
Outcome (maximum 3 points)							
Assessment of Outcome	1	1	2	2	1	1	1
Statistical Test	1	1	1	1	1	1	1
Total Points (maximum 8 points)	7	7	5	6	5	7	7

Appendix B. Adapted Newcastle Ottawa Quality Assessment Scale: Cohort Studies

Selection (maximum 4 points)

- 1) Representative: 1 point was given if the sample was truly representative of the exposed cohort (underweight or undernourished PW)
- 2) Selection of Non-Exposed Cohort: 1 point was given if the non-exposed cohort was drawn from the same community as the exposed cohort
- 3) Ascertainment of Exposure: 1 point was given ascertainment of exposure was from secure record, structured interview, or healthcare provider
- 4) Outcome of Interest at Start of Study: 1 point was given if demonstration that outcome of interest was not present at start of study

Comparability (maximum 2 points)

- 1) Adjusted for Risk Factors: 1 point was given if study was adjusted for most important/relevant risk factors
- 2) Adjusted for Additional Confounders: 1 point was given if study was adjusted for confounding factors

Outcome (maximum 3 points)

- 1) Assessment of Outcome: 1 point was given if the assessment of the outcome is satisfactory
- 2) Follow-up Time: 1 point was given if follow-up was long enough for outcomes to occur
- 3) Adequacy of Follow-Up Cohorts: 1 point was given if complete follow-up for all subjects was accounted for or subjects loss to follow-up was low

Total Points:

- 8, 9: good quality
- 6, 7: fair quality
- 5 points or less: poor quality

Study	Miele et al. 2021	Vasundhara et al. 2019
Selection (maximum 4 points)		
Representative	1	1
Selection of Non-exposed Cohort	1	1
Ascertainment of Exposure	1	1
Outcome of Interest at Start of Study	0	1
Comparability (maximum 2 points)		
Adjusted for Risk Factors	1	1
Adjusted for Additional Confounders	0	0
Outcome (maximum 3 points)		
Assessment of Outcome	1	1
Follow-Up Time	1	1
Adequacy of Follow-Up Cohorts	1	1
Total Points (maximum 9 points)	7	8

Appendix B. Adapted Newcastle Ottawa Quality Assessment Scale: Cohort Studies

Appendix C. Adapted Newcastle Ottawa Quality Assessment Scale: Case-Control Studies

Selection (maximum 4 points)

- 1) Representative: 1 point was given if the sample was truly representative of the target population (underweight or undernourished PW)
- 2) Case Definition: 1 point was given if the case definition was adequate with independent validation
- 3) Selection of Controls: 1 point was given if selection of controls was from the same source population as cases
- 4) Definition of Controls: 1 point was given if definition of controls was adequate

Comparability (maximum 2 points)

- 1) Adjusted for Risk Factors: 1 point was given if study was adjusted for most important/relevant risk factors
- 2) Adjusted for Additional Confounders: 1 point was given if study was adjusted for confounding factors

Exposure (maximum 3 points)

- 1) Ascertainment of Exposure: 1 point was given if the ascertainment of exposure was from secure record, structured interview, or healthcare provider
- 2) Method of Ascertainment: 1 point was given if method of ascertainment of exposure was the same for cases and controls
- 3) Non-Response Rate: 1 point was given if non-response rate was explained and similar for both groups

Total Points:

- 8, 9: good quality
- 6, 7: fair quality
- 5 points or less: poor quality

Study	Siyoum, Melese 2019	Bekela at al. 2020			
Selection (maximum 4 points)					
Representative	1	1			
Case Definition	1	1			
Selection of Controls	1	1			
Definition of Controls	1	1			
Comparability (maximum 2 points)					
Adjusted for Risk Factors	1	1			
Adjusted for Additional Confounders	0	0			
Exposure (maximum 3 points)					
Ascertainment of Exposure	1	1			
Method of Ascertainment	1	1			
Non-response Rate	1	1			
Total Points (maximum 9 points)	8	8			

Appendix C. Adapted Newcastle Ottawa Quality Assessment Scale: Case-Control Studies

Appendix D. Adapted Joanna Briggs Institute Critical Appraisal Checklist: Randomized Controlled Trials

- 1. Randomization: Was true randomization used for assignment of participants to treatment groups?
- 2. Allocation: Was allocation to groups concealed?
- 3. Treatment Groups at Baseline: Were treatment groups similar at the baseline?
- 4. Identical Treatment: Were treatment groups treated identically other than the intervention of interest?
- 5. Follow-Up: Was follow-up complete, and if not, were differences between groups in terms of follow-up adequately described and analyzed?
- 6. Intention-to-Treat Analysis: Were participants analyzed in the groups to which they were randomized?
- 7. Identical Outcome Measurements: Were outcomes measured in the same way for treatment groups?
- 8. Outcome Measurement Reliability: Were outcomes measured in a reliable way?
- 9. Statistical Analysis: Was appropriate statistical analysis used?
- 10. Trial Design: Was the trial design appropriate for the topic, and any deviations from the standard RCT design accounted for in the conduct and analysis?

1 point given to each question above answered with "yes", 0 points given to each question above answered with "no" or non-applicable

Total Points:

9, 10: good quality7, 8: fair quality6 points or less: poor quality

Appendix D. Adapted Joanna	Briggs Institute	Critical Appraisa	l Checklist:	Randomized
Controlled Trials				

Study	Chen et al. 2021	Callaghan- Gillespie et al. 2017	Stevens et al. 2018	Ashorn et al. 2015	Adu- Afarwuah et al. 2017
1. Randomization	1	1	1	1	1
2. Allocation	0	0	1	1	1
3. Treatment Groups at Baseline	0	1	1	1	1
4. Identical Treatment	1	1	1	0	1
5. Follow-up	1	1	1	1	1
6. Intention-to-Treat Analysis	1	1	1	1	1
7. Identical Outcome Measurements	1	1	1	1	1
8. Outcome Measurement Reliability	1	1	1	1	1
9. Statistical Analysis	1	1	1	1	1
10. Trial Design	0	1	1	1	1
Total Points (maximum 10 points)	7	9	10	9	10

Appendix E. Adapted AMSTAR Checklist: Systematic Reviews and Meta-Analyses

- 1. 'a priori': Was an 'a priori' design provided?
- 2. Duplicates: Was there duplicate study selection and data extraction?
- 3. Literature Search: Was a comprehensive literature search preformed?
- 4. Status of Publication: Was the status of publication (i.e. grey literature) used as an inclusion criteria?
- 5. List of Studies: Was a list of studies (included and excluded) provided?
- 6. Characteristics: Were the characteristics of the included studies included?
- 7. Scientific Quality: Was the scientific quality of the included studies assessed, documented, and used appropriately in formulating conclusions??
- 8. Methods: Were the methods used to combine the findings of studies appropriate?
- 9. Bias: Was the likelihood of publication bias assessed?
- 10. Conflict of Interest: Was the conflict of interest included?

1 point given to each question above answered with "yes", 0 points given to each question above answered with "no" or non-applicable

Total Points:

9, 10: good quality7, 8: fair quality6 points or less: poor quality

Appendix E. Adapted AMSTAR Checklist: Systematic Reviews and Meta-Analyses				
Study	Das et al. 2018	Oh et al. 2020		
1. 'a priori'	1	1		
2. Duplicates	1	1		
3. Literature Search	1	1		
4. Status of Publication	1	1		
5. List of Studies	1	1		
6. Characteristics	1	1		
7. Scientific Quality	1	1		
8. Methods	1	1		
9. Bias	1	1		
10. Conflict of Interest	1	1		
Total Points (maximum 10 points)	10	10		

REFERENCES

¹ Shetty P. Malnutrition and undernutrition. *Medicine*. 2003;31(4):18-22. doi:10.1383/medc.31.4.18.27958

² Malnutrition. Johns Hopkins Medicine. https://www.hopkinsmedicine.org/health/condition and-diseases/malnutrition. Published 2022.

³ Bloxham L. What's the difference between malnutrition and undernutrition, and why is it important? Concern Worldwide. https://www.concern.org.uk/news/whats-difference-between malnutrition-and-undernutrition-and-why-it-important. Published July 2020.

⁴ Dipasquale V, Cucinotta U, Romano C. Acute Malnutrition in Children: Pathophysiology, Clinical Effects and Treatment. *Nutrients*. 2020;12(8):2413. Published 2020 Aug 12. doi:10.3390/nu12082413

⁵ Ververs MT, Antierens A, Sackl A, Staderini N, Captier V. Which anthropometric indicators identify a pregnant woman as acutely malnourished and predict adverse birth outcomes in the humanitarian context?. *PLoS Curr*. 2013;5:ecurrents.dis.54a8b618c1bc031ea140e3f2934599c8.

⁶ Kpewou, DE, Poirot, E, Berger, J, et al. Maternal mid-upper arm circumference during pregnancy and linear growth among Cambodian infants during the first months of life. *Matern Child Nutr.* 2020; 16(S2):e12951.

⁷ Need to know: Muac and malnutrition. Partners In Health. https://www.pih.org/article/need-to-know-muac-and-malnutrition. Published November 7, 2019.

⁸ Weight gain during pregnancy. Centers for Disease Control and Prevention. https://www.cdc.gov/reproductivehealth/maternalinfanthealth/pregnancy-weight-gain.htm. Published May 26, 2021.

⁹ Derraik JG, Lundgren M, Cutfield WS, Ahlsson F. Maternal Height and Preterm Birth: A Study on 192,432 Swedish Women. *PLoS One*. 2016;11(4):e0154304. Published 2016 Apr 21. doi:10.1371/journal.pone.0154304

¹⁰ Wilkinson AL, Pedersen SH, Urassa M, et al. Associations between gestational anthropometry, maternal HIV, and fetal and early infancy growth in a prospective rural/semirural Tanzanian cohort, 2012-13. *BMC Pregnancy Childbirth*. 2015;15:277. Published 2015 Oct 29. doi:10.1186/s12884-015-0718-6

¹¹ World Health Organization.. *WHO recommendations on antenatal care for a positive pregnancy experience*. Geneva (Switzerland): World Health Organization; 2016.

¹² Annan RA, Webb P, Brown R. Management of moderate acute malnutrition (MAM): current knowledge and practice. *CMAM Forum Technical Brief.* Oxford (UK): Emergency Nutrition Network; 2014

¹³ Papathakis P, Schaffner A, Garcia P, Fry J, Malek S, Trehan I, Thakwalakwa C, Maleta K, Manary M. Provision of Supplementary Food to Pregnant Malawian Women with Moderate Acute Malnutrition Improves Gestational Weight Gain and Reduces Low Birth Weight. *The FASEB Journal*. Published 2018 Oct 03.

¹⁴ Stevens B, Watt K, Brimbecombe J, Clough A, Judd JA, Lindsay D. A village-matched evaluation of providing a local supplemental food during pregnancy in rural Bangladesh: a preliminary study. *BMC Pregnancy Childbirth*. 2018;18(1):286. Published 2018 Jul 4. doi:10.1186/s12884-018-1915-x

¹⁵ Schulze KJ, Mehra S, Shaikh S, et al. Antenatal Multiple Micronutrient Supplementation Compared to Iron-Folic Acid Affects Micronutrient Status but Does Not Eliminate Deficiencies in a Randomized Controlled Trial Among Pregnant Women of Rural Bangladesh. *J Nutr*. 2019;149(7):1260-1270. doi:10.1093/jn/nxz046

¹⁶ Adu-Afarwuah S, Lartey A, Okronipa H, et al. Maternal Supplementation with Small-Quantity Lipid-Based Nutrient Supplements Compared with Multiple Micronutrients, but Not with Iron and Folic Acid, Reduces the Prevalence of Low Gestational Weight Gain in Semi-Urban Ghana: A Randomized Controlled Trial. *J Nutr.* 2017;147(4):697-705. doi:10.3945/jn.116.242909

¹⁷ "Iron with or without Folic Acid Supplementation in Women." *World Health Organization*, World Health Organization, 25 Nov. 2016

¹⁸ "Corn Soy Blend/plus Commodity Fact Sheet." *Archive - U.S. Agency for International Development*, https://2012-2017.usaid.gov/what-we-do/agriculture-and-food-security/food-assistance/resources/implementation-tools/corn-soy.

¹⁹ Christian, Parul, and Keith P West. *Technical Brief - a2zproject.Org*.

²⁰ "WFP - World Food Programme Ready-to-Use Supplementary Food (RUSF)." *World Food Programme*, 31 Jan. 2021

²¹ "Balanced Energy and Protein Supplementation during Pregnancy." *World Health Organization*, World Health Organization, 11 Feb. 2019

²² de Kok B, Moore K, Jones L, et al. Home consumption of two fortified balanced energy protein supplements by pregnant women in Burkina Faso. *Matern Child Nutr*. 2021;17(3):e13134. doi:10.1111/mcn.13134

²³ Schulze, Kerry J, et al. "Antenatal Multiple Micronutrient Supplementation Compared to Iron-Folic Acid Affects Micronutrient Status but Does Not Eliminate Deficiencies in a Randomized Controlled Trial among Pregnant Women of Rural Bangladesh." *The Journal of Nutrition*, Oxford University Press, 1 July 2019,

²⁴ "WFP - World Food Programme." World Food Programme, 26 Oct. 2020,

²⁵ Sphere Association. The Sphere Handbook: Humanitarian Charter and Minimum Standards in Humanitarian Response, fourth edition, Geneva, Switzerland, 2018. www.spherestandards.org/handbook

²⁶ James, P. T., Wrottesley, S. V., Lelijveld, N., Brennan, E., Fenn, B., Menezes, R., & Mates, E. (2022). *Women's Nutrition: A summary of evidence, policy and practice including adolescent and maternal life stages*. ENN.

²⁷ Low birth weight. World Health Organization.https://www.who.int/data/nutrition/nlis/info/low-birth-weight. Published 2022.

²⁸ Peleg D, Kennedy CM, Hunter SK. Intrauterine growth restriction: identification and management. *Am Fam Physician*. 1998 Aug;58(2):453-60, 466-7. PMID: 9713399.

²⁹ Preterm birth. World Health Organization. https://www.who.int/news-room/fact-sheets/detail/preterm-birth. Published February 19, 2018.

³⁰ Schlaudecker EP, Munoz FM, Bardají A, et al. Small for gestational age: Case definition & guidelines for data collection, analysis, and presentation of maternal immunisation safety data. *Vaccine*. 2017;35(48 Pt A):6518-6528. doi:10.1016/j.vaccine.2017.01.040

³¹ Stunting in a Nutshell. World Health Organization. https://www.who.int/news/item/19-11-2015-stunting-in-a-nutshell. Published 2022.

³² Wells G, Shea B, O'Connell D, Peterson J, Welch V, Losos M, Tugwell P: The Newcastle-Ottawa Scale (NOS) for assessing the quality of nonrandomised studies in meta-analyses. 2013.

³³: Tufanaru C, Munn Z, Aromataris E, Campbell J, Hopp L. Chapter 3: Systematic reviews of effectiveness. In: Aromataris E, Munn Z (Editors). Joanna Briggs Institute Reviewer's Manual. The Joanna Briggs Institute, 2017.

³⁴ Shea BJ, Reeves BC, Wells G, Thuku M, Hamel C, Moran J, Moher D, Tugwell P, Welch V, Kristjansson E, Henry DA. AMSTAR 2: a critical appraisal tool for systematic reviews that include randomised or non-randomised studies of healthcare interventions, or both. BMJ. 2017 Sep 21;358:j4008.

³⁵ He Z, Bishwajit G, Yaya S, Cheng Z, Zou D, Zhou Y. Prevalence of low birth weight and its association with maternal body weight status in selected countries in Africa: a cross-sectional study. *BMJ Open*. 2018;8(8):e020410. Published 2018 Aug 29. doi:10.1136/bmjopen-2017-020410

³⁶ Tesfa D, Tadege M, Digssie A, Abebaw S. Intrauterine growth restriction and its associated factors in South Gondar zone hospitals, Northwest Ethiopia, 2019. *Arch Public Health*. 2020;78:89. Published 2020 Sep 29. doi:10.1186/s13690-020-00475-2

³⁷ Chen YT, Zhang T, Chen C, et al. Associations of early pregnancy BMI with adverse pregnancy outcomes and infant neurocognitive development. *Sci Rep.* 2021;11(1):3793. Published 2021 Feb 15. doi:10.1038/s41598-021-83430-7

³⁸ WHO European health information at your fingertips. 2021. https://gateway.euro.who.int/en/indicators/mn_survey_18-recommendations-based-on-prepregnancy-bmi/.

³⁹ Sultana T, Karim MN, Ahmed T, Hossain MI. Assessment of under nutrition of Bangladeshi adults using anthropometry: can body mass index be replaced by mid-upper-arm-circumference?. *PLoS One*. 2015;10(4):e0121456. Published 2015 Apr 14. doi:10.1371/journal.pone.0121456

⁴⁰ Kumar P, Sinha R, Patil N, Kumar V. Relationship between mid-upper arm circumference and BMI for identifying maternal wasting and severe wasting: a cross-sectional assessment. Public Health Nutr. 2019 Oct;22(14):2548-2552. doi: 10.1017/S1368980019000727. Epub 2019 May 14. PMID: 31084660.

⁴¹ Fakier A, Petro G, Fawcus S. Mid-upper arm circumference: A surrogate for body mass index in pregnant women. S Afr Med J. 2017;107(7):606-610. Published 2017 Jun 30. doi:10.7196/SAMJ.2017.v107i7.12255

⁴² Mishra KG, Bhatia V, Nayak R. Association between mid-upper arm circumference and body mass index in pregnant women to assess their nutritional status. J Family Med Prim Care. 2020 Jul 30;9(7):3321-3327. doi: 10.4103/jfmpc.jfmpc_57_20. Erratum in: J Family Med Prim Care. 2020 Sep 30;9(9):5085. PMID: 33102290; PMCID: PMC7567285.

⁴³ Miele MJ, Souza RT, Calderon I Preterm SAMBA study group, *et al.* Proposal of MUAC as a fast tool to monitor pregnancy nutritional status: results from a cohort study in Brazil *BMJ Open* 2021;**11:**e047463. doi: 10.1136/bmjopen-2020-047463

⁴⁴ Adane T, Dachew BA. Low birth weight and associated factors among singleton neonates born at Felege Hiwot referral hospital, North West Ethiopia. *Afr Health Sci.* 2018;18(4):1204-1213. doi:10.4314/ahs.v18i4.42

⁴⁵ Siyoum M, Melese T. Factors associated with low birth weight among babies born at Hawassa University Comprehensive Specialized Hospital, Hawassa, Ethiopia. *Ital J Pediatr.* 2019;45(1):48. Published 2019 Apr 11. doi:10.1186/s13052-019-0637-7

⁴⁶ Vasundhara, D., Hemalatha, R., Sharma, S., Ramalaxmi, B. A., Bhaskar, V., Babu, J., Kankipati Vijaya, R. K., & Mamidi, R. (2020). Maternal MUAC and fetal outcome in an Indian tertiary care hospital: A prospective observational study. *Maternal & Child Nutrition*, **16**(2), e12902.

⁴⁷ Tesfa D, Tadege M, Digssie A, Abebaw S. Intrauterine growth restriction and its associated factors in South Gondar zone hospitals, Northwest Ethiopia, 2019. *Arch Public Health*. 2020;78:89. Published 2020 Sep 29. doi:10.1186/s13690-020-00475-2

⁴⁸ Kpewou DE, Poirot E, Berger J, et al. Maternal mid-upper arm circumference during pregnancy and linear growth among Cambodian infants during the first months of life. *Matern Child Nutr*. 2020;16 Suppl 2(Suppl 2):e12951. doi:10.1111/mcn.12951

⁴⁹ Callaghan-Gillespie M, Schaffner AA, Garcia P, et al. Trial of ready-to-use supplemental food and corn-soy blend in pregnant Malawian women with moderate malnutrition: a randomized controlled clinical trial. *OUP Academic*. Published August 9, 2017.

⁵⁰ Stevens B, Watt K, Brimbecombe J, Clough A, Judd JA, Lindsay D. A village-matched evaluation of providing a local supplemental food during pregnancy in rural Bangladesh: a preliminary study. BMC Pregnancy Childbirth. 2018;18(1):286. Published 2018 Jul 4. doi:10.1186/s12884-018-1915-x

⁵¹ Bekela MB, Shimbre MS, Gebabo TF, et al. Determinants of Low Birth Weight among Newborns Delivered at Public Hospitals in Sidama Zone, South Ethiopia: Unmatched Case-Control Study. *J Pregnancy*. 2020;2020:4675701. Published 2020 Apr 16. doi:10.1155/2020/4675701

⁵² Henrick A, Imam Arundhana Thahir A, Kaluku K et al. Does Give Malnourished Pregnant Mothers with Supplementary Feeding Biscuit Can affect Pregnancy Outcomes?. *Indian Journal of Public Health Research & amp; Development*. 2020;11(1):1087. doi:10.37506/v11/i1/2020/ijphrd/193983

⁵³ Ashorn P, Alho L, Ashorn U, et al. Supplementation of Maternal Diets during Pregnancy and for 6 Months Postpartum and Infant Diets Thereafter with Small-Quantity Lipid-Based Nutrient Supplements Does Not Promote Child Growth by 18 Months of Age in Rural Malawi: A Randomized Controlled Trial. *J Nutr.* 2015;145(6):1345-1353. doi:10.3945/jn.114.207225

⁵⁴ Adu-Afarwuah S, Lartey A, Okronipa H, et al. Maternal Supplementation with Small-Quantity Lipid-Based Nutrient Supplements Compared with Multiple Micronutrients, but Not with Iron and Folic Acid, Reduces the Prevalence of Low Gestational Weight Gain in Semi-Urban Ghana: A Randomized Controlled Trial. *J Nutr*. 2017;147(4):697-705. doi:10.3945/jn.116.242909

⁵⁵ Das JK, Hoodbhoy Z, Salam RA, et al. Lipid-based nutrient supplements for maternal, birth, and infant developmental outcomes. *Cochrane Database Syst Rev.* 2018;8(8):CD012610. Published 2018 Aug 31. doi:10.1002/14651858.CD012610.pub2

⁵⁶ Oh C, Keats EC, Bhutta ZA. Vitamin and Mineral Supplementation During Pregnancy on Maternal, Birth, Child Health and Development Outcomes in Low- and Middle-Income Countries: A Systematic Review and Meta-Analysis. Nutrients. 2020;12(2):491. Published 2020 Feb 14. doi:10.3390/nu12020491z

⁵⁷ O'Dwyer V, Hogan J, Farah N, Kennelly MM, Staurt B, Turner MJ. Changes in maternal body composition during pregnancy *Archives of Disease in Childhood - Fetal and Neonatal Edition* 2012;**97:**A47.

⁵⁸ Victora, Cesar G et al. Revisiting maternal and child undernutrition in low-income and middleincome countries: variable progress towards an unfinished agenda. March 2021. The Lancet, Volume 397, Issue 10282, 1388 – 1399